Author's response to reviews

Title: Idiopathic combined, autoantibody-mediated ADAMTS13-factor H deficiency in TTP-HUS

Authors:

Daniel Patschan (d.patschan@gmail.com)
Peter Korsten (peter.korsten@web.de)
Arne Behlau (Arnea.behlau@web.de)
Radovan Vasko (rvasko@inbox.com)
Malte Heeg (malte.heeg@med.uni-goettingen.de)
Nadera Sweiss (nsweiss@medicine.bsd.uchicago.edu)
Gerhard A Müller (nephrorheuma@med.uni-goettingen.de)
Michael Koziolek (mkoziolek@med.uni-goettingen.de)

Version: 2 Date: 6 August 2011

Author's response to reviews: see over
Dear Prof. Kidd,

Herewith we would like to submit a case report with the title:

**Idiopathic combined, autoantibody-mediated ADAMTS13/factor H deficiency in TTP-HUS**

The data has never been published before and at the moment it’s not under consideration for publication elsewhere. The data presented by us will substantially broaden the current knowledge of thrombotic microangiopathies since the defect which we identified in the patient has never been reported before.

Since the patient is a minor, we have obtained written informed consent from the patient’s legal guardian for publication of the case.

Sincerely,

D. Patschan, M.D.